Author's response to reviews

Title: Factors associated with patients self-reported adherence to prescribed physical activity in routine primary health care

Authors:

Matti E Leijon (matti.leijon@med.lu.se)
Preben Bendtsen (preben.bendtsen@liu.se)
Agneta Ståhle (agneta.stahle@ki.se)
Kerstin Ekberg (kerstin.ekberg@liu.se)
Karin Festin (Karin.festin@liu.se)
Per Nilsen (per.nilsen@liu.se)

Version: 4 Date: 12 May 2010

Author's response to reviews:

To the editors, BMC Family Practice. Malmö 12 May 2010

Dear Madame/Sir,

According to our manuscript, 'Factors associated with patients self-reported adherence to prescribed physical activity in routine primary health care'

We are grateful to the editor and the reviewers for your relevant and helpful questions and comments. We are sorry about missing to remove the reference in the method section (sometimes tricky when using Endnote!). Due to your question concerning ethical approval: this data collection and follow-up was part of the routine health care as part of ordinary quality work. Therefore we believe that our statement is correct.

We would be happy to supply any additional information you require.

Yours Sincerely,

Matti E Leijon